(2020) 8:37

### **RESEARCH ARTICLE**



# Experiences of patients with anorexia nervosa during the transition from child and adolescent mental health services to adult mental health services



Veronica Lockertsen<sup>1,2\*</sup>, Liv Nilsen<sup>1</sup>, Lill Ann Wellhaven Holm<sup>3</sup>, Øyvind Rø<sup>4,2</sup>, Linn May Burger<sup>5</sup> and Jan Ivar Røssberg<sup>1,2</sup>

### Abstract

**Background:** The transition between the Child and Adolescent Mental Health Services (CAMHS) and the Adult Mental Health Services (AMHS) is identified as an especially critical time for the continuity of care for patients with anorexia nervosa (AN). However, research on this topic is scarce. In the present study, we explore the patients' experiences of the transition between CAMHS and AMHS.

**Methods:** A qualitative explorative study was carried out based on recorded interviews from one multi-step focus group and six individual interviews with patients who experienced the transition from CAMHS to AHMS in Norway. This study is service user-initiated, meaning service users were involved in all steps of the research process.

**Results:** The adolescents' experiences are characterized by four overall themes regarding the transition process between CAMHS and AMHS: (1) "Being unprepared and alone in the transition process" describes how a lack of preparation for the transition between CAMHS and AMHS makes them feel alone and increases stress. (2) "It takes time to create a trusting relationship" describes how time influences patients' trust in therapists and motivation for treatment. (3) "We are not all the same" describes how adolescents develop differently but are not treated differently despite their diverse ability to be self-sufficient. (4) "How they see me and treat me affects my hope for the future" describes the interaction between adolescents and therapists.

**Conclusions:** Acknowledging the patients' needs during the transition period and considering their readiness for the transition is important. Taking into account the four dimensions described in the present study might improve the transition process and enhance the patients' self-sufficiency and maturity.

Keywords: Anorexia nervosa, Mental health transition, Adolescent, Patients' perspective, Qualitative research

\* Correspondence: v.lockertsen@gmail.com

<sup>1</sup>Division of Mental Health and Addiction, Oslo University Hospital, P.O. Box 4959, Nydalen, Oslo, Norway

<sup>2</sup>Institute of Clinical Medicine, Faculty of Medicine, University of Oslo, 0318 Oslo, Norway

Full list of author information is available at the end of the article



<sup>©</sup> The Author(s). 2020 **Open Access** This article is licensed under a Creative Commons Attribution 4.0 International License, which permits use, sharing, adaptation, distribution and reproduction in any medium or format, as long as you give appropriate credit to the original author(s) and the source, provide a link to the Creative Commons licence, and indicate if changes were made. The images or other third party material in this article are included in the article's Creative Commons licence, unless indicated otherwise in a credit line to the material. If material is not included in the article's Creative Commons licence and your intended use is not permitted by statutory regulation or exceeds the permitted use, you will need to obtain permission directly from the copyright holder. To view a copy of this licence, visit http://creativecommons.org/licenses/by/4.0/. The Creative Commons Public Domain Dedication waiver (http://creativecommons.org/publicdomain/zero/1.0/) applies to the data made available in this article, unless otherwise stated in a credit line to the data.

### **Plain English summary**

Adolescents with anorexia nervosa (AN) who receive care from Child and Adolescent Mental Health Services (CAHMS) often need further treatment from Adult Mental Health Services (AHMS). This study explored the transition between CAHMS and AHMS for patients with AN in a naturalistic public mental health-care setting. Our study identified four themes concerning the transition between CAMHS and AMHS, which patients considered influenced their transition and which demonstrated their lack of preparedness for the transition: The patients often felt alone in this period; time was considered an important element in developing trust in the therapeutic relationship; having the transition adapted to their individual needs and developmental stage is important for the transition outcome; how they were introduced to their new therapists and treated had an effect on their hope for recovery.

### Introduction

As many patients experience the transition process between Child and Adolescent Mental Health Services (CAMHS) and Adult Mental Health Services (AMHS) as less than optimal, and many patients drop out of treatment [1], continuity of care in the transition from CAHMS to AMHS is of major importance. A qualitative meta-synthesis focusing on the transition of chronically ill adolescents found that adolescents are unprepared for the transition to AMHS and that the loss of a familiar surroundings and relationships they had at CAMHS creates anxiety [2]. Adolescents are vulnerable to the changes that occur during the transition process, and the inflexibility of the service systems puts many adolescents at risk of disengaging from further treatment [1]. Less than 5% of 154 adolescents with a mental health disorder experienced a satisfactory transition [3]. Research show that several mental health care systems, do not have satisfactory cooperation between the CAMHS and AMHS, which creates a lack of continuity for mental health patients [4, 5]. It is thus recommended that the transition process from CAMHS to AMHS should be carefully planned, taking into consideration the medical, psychosocial, and educational/vocational needs of the adolescents [6, 7]. The patients should be properly engaged at AMHS before their therapy at CAMHS is terminated, and they should view the transition period as a time to practice self-sufficiency and increase empowerment. A strong focus on family involvement in the transition period, is important [8, 9]. Anorexia nervosa (AN) is a severe psychiatric disorder that patients often struggle with for several years. The disorder often results in severe psychosocial and physical consequences for the individual and their family. Age of onset is typically between 14 and 19 years [10], and approximately 85% of

the cases begin before the age of 20 [11]. Additionally, patients with AN often have comorbid disorders, such as substance abuse, depression, obsessive-compulsive disorder (OCD), anxiety, and post traumatic stress disorder (PTSD), which complicate the recovery process [12, 13]. Patients with AN often have a devastating fear of gaining weight; the nature of an eating disorder often implies that the patient does not consider themselves ill and has poor insights into the consequences of the illness, which frequently creates a struggle with motivation for recovery [14]. Patients claim that AN gives them a feeling of control and protection, making it difficult for them to let go of the disease and thus comply with treatment. As a result of the above, patients with AN often have complex needs in the transition period, and a thoughtful transition plan is necessary to maintain treatment progress [15]. Although there are variations between different local treatment cultures, CAMHS and AMHS have quite different approaches regarding the extent to which they engage the parents in the treatment [16]. The treatment approach at AMHS is based on individual responsibility. Dimitropoulos, Toulany [17] questioned these sudden changes in treatment approaches to determine whether the supporting surroundings at CAMHS offer adolescents the opportunities they need to practice self-sufficiency and eat independently without familial support. As patients with AN are thought to lag behind their peers both developmentally and psychologically, they typically require careful preparation to manage the transition process [18]. Professionals have underlined how a barrier is formed in the transition when age, rather than readiness, is set as a marker for timing the transition [18]. Most high-income countries divide their services between CAMHS and AMHS at around age 18 [19].

To increase the possibility of adolescents managing the transition, it is essential to have good collaboration between the two services. Nevertheless, a previous study found that the current lack of cooperation between CAMHS and AMHS makes it difficult for patients to navigate between the two services [20]. This gap seem to be present in several countries, regardless of how CAMHS and AMHS are organized [21]. There have been an increasingly awareness of the challenge's patients have in the transition process between CAMHS and AMHS. Based on equal input from relatives, professionals, and the patients, the Canadian Eating Disorder Priority Setting Partnership has prioritized research topics for female patients with AN [22]; they identified that one of the top priorities relates to gaps in treatment, suggesting that research should focus on the transition between services. "Integrated care" have also been identified by clinicians and service users, as a research priority in an Australian setting [23]. Although previous studies have investigated relatives', professionals', and patients'

experiences of transitions in specialized eating disorder programs [17, 18, 24], few have focused solely on the patients' perspectives during this transition period, when treated in a naturalist public mental health-care setting. A greater understanding of how patients experience this transition through the different levels of organized health care could contribute to a more cohesive treatment process for patients. Thus, we interviewed patients with AN to understand their experiences of the transition from CAMHS to AMHS.

### Methods

### Design

The current study is part of a qualitative research project designed to investigate the experiences of the transition between CAMHS and AMHS for patients, parents, and professionals called Mind the Gap. To investigate the transition between CAMHS and AMHS from different perspectives, we used qualitative techniques, including semi-structured interviews and multi-step focus groups [25]. Mind the Gap is a user-initiated project, which implies that the experiences from a support group for relatives form the basis of the research questions. In the current study, the involvement of service users have influenced the research process from the start (designing the study) to the final version of the present manuscript. A former service user who has experience with AN (LMB) participated in the design of a semi-structured interview guide and moderated the interview in collaboration with the first author (VL). The research group also included a former service user with experience as a relative to a patient with AN (LAWH), two RN nurses (LN, VL), and two psychiatrists (JIR, ØR). We found this multi-perspective research group effective for forming the research questions, designing the interview guide, preparing and conducting the focus group interviews, and analyzing the transcriptions.

### **Recruitment and participants**

A total of 10 participants were included in the study. All participants had experiences with the transition between CAMHS and AMHS in a specialized mental health care services. Six had experiences from inpatient units and three had been treated in a specialized eating disorder unit in CAMHS. Five of the participants had been treated in a specialized eating disorder unit in AMHS. All participants were women, with a median age of 22 (range; 19–29) years. The median time since the transition between CAMHS and AMHS was 4.5 years (range; 1–10 years). Six of the participants were working or studying at the time of the qualitative interview. Eight of the participants were invited to participate by current therapists who contacted participants they knew had experiences

that could illuminate the research questions. The others made contact with the project manager after finding project information on an internet support site for eating disorders. Recruitment took place between February 2018 and May 2019. Written consent was obtained from all participants.

### Data collection

Semi-structured interviews (lasting 60-90 min) were conducted by the fifth (LMB) and first (VL) authors as moderator and co-moderator, respectively. In addition to former user experiences, LMB has attended service user boards in hospitals and performed voluntary work in an interest group for eating disorders. We conducted five individual interviews and one multi-step focus group interview with four participants to explore the experiences of AN patients in the transition between CAMHS and AMHS. The individual and focus group interviews were both guided by our research question, and participants were encouraged to provide concrete examples of their experiences. Focus group interviews are described as a suitable technique in qualitative research because they elicit people's understandings and views of their experiences [26]. Focus groups can be stimulating environments, with co-participants positively triggering other participants' memories. Multi-step focus groups are considered a beneficial method when working with service users in research [27]. Repeated meetings give the participants an opportunity to inspect and challenge their own and others' experiences in a dialectic way. Between the focus group meetings, we handed out a summary of the first focus group interview, so the participants could prepare for the second meeting. This summary constructed the main themes we wanted to focus on in the second group meeting. The individual interviews supplied an in-depth answer of the research question. The first author recorded and transcribed the interviews verbatim. The interviews were conducted at the research clinic or in other neutral settings.

### Data analysis

Data were analyzed using a systematic text condensation inspired by Giorgi [28, 29]. The method focused on presenting the patients' expressed experiences, rather than a possible underlying meaning of what was said. Using a method of de-contextualization and re-contextualization of the text, we analytically reduced the data and created a condensed text that represented the overall material [28]. The process comprised four steps: In the first step, the research group read the interviews several times to form an overall impression of the material. In step two, we started sorting the text. Text that belonged together was grouped and encoded. The codes were adjusted and organized so that themes with common content were collected under the same code. In step three, the meaning contents of the different codes were abstracted and collected across the interviews. Step four is the analytical text that represents the results of the research project. With our co-operative relationship, we managed to have a multi-perspective view of the data. To help organize and categorize the transcriptions, we used the computerized program NVivo 11 (QSR International Pty Ltd.)

### **Ethical aspects**

The study was approved by the data protection authority at Oslo University Hospital OUS (2016/19732) and performed in accordance with the Declaration of Helsinki. The Regional Committee for Medical and Health Research Ethics for the Southeast Region of Norway (2016/ 1259) viewed the study not to be a medical or healthrelated research project regulated by the law of health research and was not therefore subject to presentation, cf. HFL. § 2.

### Results

The participants' experiences are characterized by four overall themes regarding the transition process between CAMHS and AMHS: (1) "Being unprepared and alone in the transition process" describes how a lack of preparation for the transition between CAMHS and AMHS makes them feel alone and increases stress. (2) "It takes time to create a trusting relationship" describes how time influences patients' trust in therapists and motivation for treatment. (3) "We are not all the same" describes how participants develop differently but are not treated differently despite their diverse ability to be self-sufficient. (4) "How they see me and treat me affects my hope for the future" describes the interaction between participants and therapists.

### Being unprepared and alone in the transition process

The participants described a feeling of being unprepared and alone during the transition between CAMHS and AMHS. Some patients described that their Primary Care Doctor (PCP) tried to compensate for the lack of followup by offering ad hoc support: "I went so many months on my own-without professional contact, just with a support-call from my PCP doctor." They felt overwhelmed when entering AMHS and were unprepared to handle the differences between CAMHS and AMHS. When treated at CAMHS, they felt they were followed up more closely during unstable periods. However, at AMHS, they were often discharged: "Here I got discharged when I was too ill to manage the treatment program." The participants believed they would feel less alone and stressed if CAMHS had a more active function in the transition process. "I remember my mom called CAMHS to ask if I could come back, just after turning 18, but they turned us down." Some participants explained that their parents became increasingly discouraged because they felt as if they had nowhere to turn to for help: "My parents lost hope after CAMHS because they thought that CAMHS had tried everything and that we had nowhere to turn to. So, my parents believed that this would last for the rest of our lives." Often when the participants felt alone and scared during the transition process, they turned to and focused more on their eating disorder to gain a sense of control in the situation. However, focusing on their AN ameliorated their internal chaos. Most of the participants said they were unprepared for the expectation at AMHS for them to be selfsufficient and for how their parent's role changed during the transition phase: "I went from being admitted with my parents in CAMHS to being transferred to a ward that treated my parents as distant relatives, with restrictions about when to call them and what to let them know." It was new for the patients to be responsible for passing information forward and managing the collaboration with their parents. Following the active role their parents had played in CAMHS, some participants felt it important to establish themselves as independent and maintain control. Therefore, many participants found it difficult to uphold an open dialog with their parents during the transition period. Differences in how parents were included in the treatment between CAMHS and AMHS made the patients concerned about their parents' well-being. They considered it important that their parents also received support and help from the therapists at AMHS.

### It takes time to create a trusting relationship

Overall, the participants viewed their first meeting at AMHS as important for their adherence to the treatment. The participants often described themselves as being insecure when establishing a relationship at AMHS. They felt they were met with a preconception of what they were, and they were afraid that the new therapists would be unable to understand and acknowledge their situation. The participants were worried about being misinterpreted and were concerned that their story could turn into something they did not recognize as their own. Consequently, they held back, but they acknowledged that doing so made it difficult for the new therapist to help them. The participants stated that it would have been easier for them to connect if CAMHS had prepared them for the transition. Some participants believed that it would be easier to communicate with their new therapist at AMHS if their CAMHS therapist had been present for support in the first meetings. As one adolescent said, "I got scared in the first meeting because I was used to a warmer, more caring, and humble approach. At CAMHS, they often sat beside me, but at AMHS, I sat on the opposite side of a large table with a therapist that did not say much." When establishing a new relationship, the participants needed time to open up about their weight and food issues. They expected the therapists to understand this need, so it was difficult for the participants to understand why they rushed them into gaining weight, and why they-so early in the relationship-gave the responsibility for recovery back to the patients. They felt a need for more structure and time to feel safe and ready to work on their problems regarding food and their emotions. The participants often described how difficult it was to engage in treatment when entering AMHS. They mostly explained this in terms of how the treatment was carried out at CAMHS and how it ended. As they were defined as adults when they turned 18, CAMHS terminated the therapy. While some understood why the transition was timed as it was and felt included and ready for a change; others felt the timing to be arbitrary and felt ignored in the transition process. This made them feel let down by CAMHS, which led some to mistrust towards the treatment system.

The participants felt that the therapists failed to understand what state they were in and how long it could take to build a satisfactory relationship between them and the therapists; they expected too much, too fast. For the participants, this created an issue of mistrust and allowed room for misunderstandings between the therapist and patients: "Perhaps she just wanted to tell me that this is going to be all right, but that was not the way I understood it."

During the transition process to AMHS, the patients described many insecurities about who they were going to meet and how they would be met. They described how they were moved from person to person and how they were expected to follow and trust each new person they met. This was challenging for the participants because the information between CAMHS and AMHS was lacking, and the participants often had to repeat themselves, which made it difficult for the participants to develop motivation for their recovery. However, they managed to gain weight just by doing what they were told: "I realized that I needed the help-but I was never safe enough to choose to stay voluntary." The participants said it was important to have a therapist who showed that they cared about them and their problems. To find a way to trust their therapist, the participants described different ways to test the therapists' persistence and motivation. It could be as specific as in a meal situation or in evaluating the therapists' responses in conversations, e.g., what words they used. If therapists showed that they cared, and "passed their tests," the patients described that the motivation for recovery often increased, which gave them hope:

It was really important for me to have them close and see their engagement. That was the first time I had my own desire for recovery. You get so tired of the fight between your own two voices that it has to be the people meeting you that figure it out.

The participants described how much easier it was to trust therapists that seemed to be transparent. Some of the patients had experiences with therapists at CAMHS who they felt misused their trust by hiding information from them and going behind their backs to talk to their parents. When they had met a therapist had helped them, they found that they were often the ones who had spent the most time with them and were authentic:

It takes more time than a biweekly session. Now is the first time I have had a relationship with a therapist in almost a year. It is working now because we have passed the point of meeting each other with our assumptions, and I know he knows me—for me. I can be honest, and he knows my little signs of hurting.

### We are not all the same

Some of the participants felt ready for the responsibility that AMHS expected of them, which was to based on AMHS's more individual, independent approach. Others found this shift in approach from that used at CAMHS too demanding. The participants missed being met with curiosity about what they viewed best for themselves during the transition process, including their ideas about how to recover. Despite having the same diagnosis, the participants were in different developmental stages and thus had different needs during the transition: "I was not ready-they wrote from CAMHS that I was eighteen on paper, but my mind was fourteen." The participants described a sense of lagging behind their peers because they had been ill for many years. Some participants explained that they felt almost developmentally disturbed because the years during which they lacked normal contact with school and friends had had a fundamental effect on their development and social skills. It can be painful being defined as an adult when you feel much younger. They felt that the mental health system has a rigid understanding of what an adult is, being defined by the age boundary of 18: "They have to look into the situation and the person. We are not the same." The participants agreed that the transition should be timed based on their developmental stage rather than their age.

While some described the transition to AMHS as something that prolonged their recovery process, others described it as vital for their improvement. They claimed that they were somewhat tired of CAMHS's more childish approach and were ready for the transfer at the age of 17. They felt they were emerging into adulthood and were treated with too much caution at CAMHS: "I had to solve some problems myself and could not depend on other grownups. I had to be a person in my own life." Although some of the participants had negative experiences of the transition, they thought it was important for their recovery process because it underlined the changes they had to make in regard to how they take responsibility. They claimed that "the transition can help reinforce that you are the most important person in your life and the only one who can change something. You're your mom or your therapist." They began to realize that they had to change, take control of their life, and become accountable to themselves, but they had to be ready to do so.

### How they see me and treat me affects my hope for the future

The participants often underlined that they felt powerless during the transition; they were dependent on how the system and professionals related to them and perceived them during the transition process. These interactions not only had a practical meaning, but they were also deeply important to the participants. The participants described how the choice of words, spoken or written, influenced their motivation for recovery: "When you are met with 'this is chronic' at the age of 17, what is the point of trying? This just confirms that it is hopeless." For the adolescent patient, it was difficult to maintain motivation when they sensed that their surroundings did not envision their recovery. According to the participants, the difference between CAMHS and AMHS was evident in how they verbalized their visions and hope for the adolescent patient. While CAMHS focused on their future and the life they had in front of them, giving them hope, AMHS did not have the same focus in, for example, how they facilitated maintaining obligations to school or work. The participants perceived an attitude at AMHS that suggested they had to figure those things out on their own. Some participants were transferred from CAMHS to an adult acute psychiatric mental health ward, which they found particularly challenging. They described their mental state as being in crisis and could not understand why CAMHS refused to prolong their treatment. They thought that AMHS did not understand how frightening it could be to be admitted in an adult environment:

I had really nothing to say in that decision; they just let me know that I was going. I stayed for 8 months. I do not think that a 17–18-year-old is suited to an adult acute mental health care unit."

The powerlessness they felt during the transition underlined feelings of aloneness and being unvalued. They describe a sense of resignation—"so I lost interest"—mostly because they felt AMHS lacked an understanding of the difficulties they were experiencing during the transition period.

When treated in in-patient care, they were admitted with other patients who had been ill for a long time. These factors influenced the participants' hope for their recovery in the future: "At AMHS, I was admitted with patients that had been ill longer than I had been living." They described a system that met them with phrases like "early disability benefits," and their illness was described as "treatment resistant." They explained how such words affected their views of their future because such words would stick, influencing how the rest of the treatment system viewed them and contributing to misconceptions, which narrowed their room to change. As a result, the participants became more dependent on their relationships outside of the treatment system to maintain hope for them, as they found it challenging on their own.

When entering the transition phase, the participants often felt as though AMHS had more interest in their illness than in them. The therapist seemed to place more importance on evaluating and interpreting their symptoms and BMI than talking with them about their wellbeing and experiences of their condition. Some stated that everything in therapy at AMHS was oriented around food and weight, which made it difficult to create trusting relationships with the therapists, who interpreted their BMI as mental and emotional problems. There was a high focus on their somatic health, but little interest in their mental health: "If you eat your food, then you are OK. If your BMI is close to normal, you are good." They were asked if they wanted to have their parents with them, but when they responded negatively, they were never asked "why not." The participants felt that therapists at CAMHS seemed more interested in getting to know them as a person, while those at AMHS only wanted to understand their disease: "You become your symptoms and they just ask: how can we medicate that?" With the experienced alteration from being treated as an individual to being treated as an illness, the participants often felt like an object during the transition, dependent wholly on a relationship that was restricted more to the registration of symptoms and less to what they felt would help them. They verbalized a sense of no longer feeling like a human, just classified symptoms with associated expectations about what they should be struggling with. In their adolescent years, many of them only had social contact with people who were getting paid to be there. They had to protect themselves from becoming attached to the people surrounding them, which they felt changed how they related to people in their lives outside the mental health services. For some, that situation magnified the symptoms of the AN because of the safety they represented:

In the beginning, I got really attached to people, and when they got replaced, I was really affected. But now, it affects me less and less, and the AN has become more stable and safer—and it is always there for me.

### Discussion

The findings of the current study provide a meaningful contribution to the research, which explores the experiences of participants with AN during the transition between CAMHS and AMHS. The qualitative analysis revealed four overall themes regarding the transition process: (1) "Being unprepared and alone in the transition process," (2) "It takes time to create a trusting relationship," (3) "We are not all the same," and (4) "How they see me and treat me affects my hope for the future." The themes are interrelated and display elements with the transition that positively and negatively influence the participants' experiences. These essential themes are discussed in the following paragraphs.

### Being unprepared and alone in the transition process

The current study found that the participants were unprepared for the transition to AMHS, which led to increased feelings of being alone. Without an understanding of the differences in the treatment approaches between the services, they did not feel as though they were taken seriously at AMHS. They felt alone and without what they considered adequate support during the transition phase. Our study supports the findings of Newlove-Delgado, Ford [30], who found that participants with ADHD were unprepared and left alone during the transition. As in our study, only one of the seven participants included in the study experienced to be prepared for the transition. Good preparation was followed by less stress and fewer feelings of uncertainty. Exploring the transition from the perspectives of 21 participants treated by the mental health services, Cleverley, Lenters [31] described the transition as "objectively terrifying" and the lack of preparation as "overwhelmingly present." In the current study, it is unclear whether the experienced cultural and philosophical differences in approaches, was uncommunicated or not understood by the participants. Nonetheless, Winston, Paul [16] underlined the importance of discussing the differences between the services explicitly, with both the participants and their families. Consequently, the participants in current study lacked the ability to handle the sudden expectations of autonomy. Their destructive eating routine increased when in need for a feeling of safety they felt the illness (AN) provided them when the services did not. When exploring the transition of the patients with AN from the therapists' perspectives, Dimitropoulos, Tran [14] emphasized that for patients with AN, preparing for the transition is particularly important because they often display less self-sufficiency and have difficulties fully engaging in treatment. Similar to our study, Dimitropoulos, Toulany [17] found that it would be beneficial for the patients and their relatives to have a clearer understanding of the implications of the treatment methods used in AMHS. Since patients often are dependent on parents' external support in this phase, Dimitropoulos, Tran [14] underlined the importance of relatives being involved in the process. It could be argued that with more information and better preparation, parents would be better able to provide the support participants need during the transition process. As such, adequate support and preparation from the care system could decrease the patient's feelings of being alone during this time. Ideally, it should be a period of shared care between CAMHS and AMHS, giving the patients the opportunity to understand the way the adult team works to properly prepare them for the change.

### It takes time to create a trusting relationship

In the current study, some patients had difficulty starting up a new relationship at AMHS with an open and trusting mind, and they supposed it would have been easier if they had received more support from CAMHS during the transition. The therapists' approach was considered vital when establishing a therapeutic alliance as the patients' social communication became more difficult when they felt unsafe. In accordance with our findings, previous studies have demonstrated that patients with AN are often particularly sensitive to other people's opinions about them [32]. Similarly, Sly, Morgan [33] found that the patients' first impressions of their new therapist often dictated if they would be open about themselves. They also found that young patients with AN needed a therapist that "talk[ed] straight" and required patients to be honest and open about their eating disordered behaviors; otherwise, the patients would disengage from the treatment.

The findings of this study underline the importance of the therapist's timing to address sensitive subjects. The current study found that the participants wanted their therapists to take more time to get to know them before talking about the elements that triggered them most. A major focus on food, weight, and body from the start of the therapeutic relation made it difficult to trust their therapist. Previous research has underlined the importance of the therapist's willingness to move slowly and collaborate on the therapeutic goals with the patients, especially in the earlier stages of therapy [34]. A trusting therapeutic relationship with patients with AN is considered a vital component of successful treatment, as these patients often are difficult to engage in treatment [16]. The participants often described how their ambivalent feelings toward recovery made it problematic to establish a new relationship with the therapists at AMHS. Ambivalence is described as one of the most challenging features of AN [35]. Swift and Greenberg [36]

summarized 669 studies on dropout, and they found that studies with eating disorders, personality disorders, and higher proportions of young patients had higher dropout rates than other disorders. A stronger focus on building an alliance with the patient in these first meetings would influence the patients' experience and their compliance with treatment. One issue that emerges from these findings is the complexity of clinical settings. Although the therapists acknowledge that establishing an alliance with patients is essential, they also have to meet expectations concerning efficiency, which gives them little time to do so [20].

### We are not all the same

In our study, the participants experienced the transition as an inflexible process. Some of the participants did not feel ready to be transferred to AMHS, but they had reached the age limit and were transferred anyway, which made them feel powerless. The participants viewed their developmental stage as more important for a successful transition than age. This is in line with findings of other studies conducted on transitions between CAMHS and AMHS [9, 17, 31] and in contrast to the clinical guidelines that guide therapists to collaborate with patients and relatives to find the optimal time for the transition [37]. The present study describes how the participants felt as though they were treated more as a patient than an individual when not included in the decisions made about their transition. This experience contrasts with the idea of the transition being a therapeutic process that empowers the patient's recovery process [38].

As in our study, Cleverley, Lenters [31] underlined how the participants already are in a stressful situation and in need of a more individually adapted transition. Many participants thought that the transition would be better in the early 20s, when their lives were more stable. These results reflect some of the discussion about the term "adolescence." The United Nations defines adolescence as the period of 10-19 years of age; however, Sawyer, Azzopardi [39] argued that this definition is outdated when considering contemporary patterns. They consider that extending the age group to 24 years corresponds more closely to adolescent growth and a contemporary understanding of this life phase. These ideas are consistent with other emerging literature on mental health transitions [40-42]. Raballo, Poletti [43] are concerned about the unwillingness to change the system to a more patient – oriented care that incorporates the full continuum of service response within a promotion and prevention framework for youth mental health. They are describing a more radical approach and suggest a substantial framework shift that reorient existing services towards youth- specific mental health care. With their proposition, the adolescent's developmental stage and readiness would determine the timing of the transition. We assume that with such an approach the negative experiences described in the current study, would have been reduced.

## How they see me and treat me affects my hope for the future

The participants often described how they felt powerless, dependent on the system and the therapist's impression of them, and were repeatedly defined as an "anorectic patient." To maintain their motivation to recover, the participants needed to experience a therapist that perceived them as a person with unique qualities. Earlier research has described how patients with AN often feel treated as a diagnosis [44, 45]. Malson, Finn [46] found that both the patients and the therapists construct "the anorexic patient." Defined by their disorder, everything the "patient" says is part of the AN, not the person. This resonates with our study, where participants expressed that this objectification of themselves influenced not only their self-worth negatively but also often their commitment to the treatment. These results reflect earlier studies' claims that patients with AN associate negative treatment experiences with being treated as an anorexic, rather than an individual [47, 48].

One important finding in our study was how the participants emphasized the therapists' use of words that deprived them of motivation and hope for a better future. Words such as "chronic" and "treatment resistant" made them lose hope for further treatment progress and the future. In a previous study, the interviewed professionals treating patients with AN were aware of how patients appraised and interpreted words about themselves or their condition. The professionals thus set high standards for how they communicated with the participants [20], and they felt under high pressure to do so from the patients and themselves. They also described a high level of anxiety and uncertainty concerning their assessments, and they elaborated on the consequences of potential premature termination of treatment. Empirical research suggests that therapists tend to develop more feelings of frustration, hopelessness, and helplessness in therapy with AN patients than with patients with other eating disorders [49]. However, existing studies tend to focus on patients' characteristics when addressing dropout from treatment and fail to include systemic or therapist factors, implying that the high dropout rates are primarily the patients' responsibility [36, 50]. Some issues that emerged from the findings of our study relate to the rapid turnover of therapists and how this affected how they attach themselves in relationships with their therapists and in private relationships. Swift and Greenberg [36] examined the premature discontinuation of adult psychotherapy and suggested that future therapeutic

relationships can be seriously impeded by a repeated loss of therapists. One hypothesis is that when patients are forced into new relationships that rapidly terminate, it has a negative influence on their own and their therapist's sense of achievement. This amplifies negatively both parties' abilities to create a therapeutic environment and form a basis for a future trusting relationship.

### Strengths and limitations

The inclusion of experiences from a heterogenic patient group and the multi-perspective approach when analyzing the material may have provided validity to the study, otherwise missed [51-53]. This study has several limitations. The participants were a mix of self-selected and purposively selected participants, selected by a therapist or former therapist who viewed them as suitable for participating in the current study. The present study included only female participants; thus, a potentially different selection of participants may have emphasized other experiences. One obvious limitation is the retrospective nature of the interviews, which could represent a bias in material, as one often tends to remember the negative aspects better than the positive. The timespan of 0-10years is wide; thus, participants could have been subject to recall biases. However, our study also had a thematic positive element. The participants connected meaningful elements from the transition to descriptions of consequences of the transition. Conducting qualitative interviews, the context as a phenomenon may influence the findings, here highlighted in two different ways. When trying to understand why and how particular experiences come into being, it can be a question of contextual factors. As we focus on patients' experience, these are generated from different contexts and from different parts of the country and treatment facilities [54]. The interviews in the current study were conducted in different contexts and the researchers held different roles in each interview. It is likely to assume the researcher's role as more active in the individual interviews compared to the focus group interviews. This, may have affected the contents of the interviews. At the same time, conducting individual interviews might also empower the participants by increasing their time frames to convey personal experiences without being influenced or modified by fellow participants.

### Conclusion

To conclude, it is of major importance to acknowledge the patients' needs during the transition period and considering their readiness for the transition. To better the chances of a successful transition, both patient and relatives should be explained, in an explicit manner, the differences between the services. A stronger focus on building an alliance with the patient in the first meetings would influence the patients' experience and their compliance with treatment. Taking into account the four dimensions described in the present study might improve the transition process and enhance the patients' self-sufficiency and maturity.

### Abbreviations

CAMHS: Child and adolescent mental health services; AMHS: Adult mental health services; AN: Anorexia Nervosa; OCD: Obsessive-compulsive disorder; PTSD: Post traumatic stress disorder; PCP: Primary Care Doctor

### Acknowledgements

Thank you to all participants who supported this research study.

### Authors' contributions

Design of the Mind the Gap study: LN, JIR, and LAWH. Data collection: VL and JIR. Drafting the article: VL, JIR, and LN. All authors contributed to data analysis, interpretation, and critical revision of the article as well as final approval of the version to be published.

### Funding

This project has been made possible by the Dam foundation.

#### Availability of data and materials

The datasets used and/or analyzed during the current study are available from the corresponding author upon reasonable request.

### Ethics approval and consent to participate

The study was approved by the data protection authority at Oslo University Hospital (OUS; 2016/19732). The study was performed in accordance with the Declaration of Helsinki and evaluated by the Regional Committee for Medical and Health Research Ethics for the Southeast Region of Norway (2016/1259). The study is not viewed to be a medical or health related research project regulated by the law of health research and is thereby not subject to presentation (cf. HFL, § 2).

### Consent for publication

Not applicable.

### **Competing interests**

The authors declare that they have no competing interests.

#### Author details

<sup>1</sup>Division of Mental Health and Addiction, Oslo University Hospital, P.O. Box 4959, Nydalen, Oslo, Norway. <sup>2</sup>Institute of Clinical Medicine, Faculty of Medicine, University of Oslo, 0318 Oslo, Norway. <sup>3</sup>Oslo, Norway. <sup>4</sup>Regional Department for Eating Disorders, Division of Mental Health and Addiction, Oslo University Hospital, Ullevål HF, Postboks 4950 Nydalen, 0424 Oslo, Norway. <sup>5</sup>Drammen, Norway.

### Received: 12 May 2020 Accepted: 12 July 2020 Published online: 11 August 2020

### References

- Broad KL, Sandhu VK, Sunderji N, Charach A. Youth experiences of transition from child mental health services to adult mental health services: a qualitative thematic synthesis. BMC Psychiatry. 2017;17(1):380.
- Fegran L, Hall EO, Uhrenfeldt L, Aagaard H, Ludvigsen MS. Adolescents' and young adults' transition experiences when transferring from paediatric to adult care: a qualitative metasynthesis. Int J Nurs Stud. 2014;51(1):123–35.
- Singh SP, Paul M, Ford T, Kramer T, Weaver T, McLaren S, et al. Process, outcome and experience of transition from child to adult mental healthcare: multiperspective study. Br J Psychiatry. 2010;197(4):305–12.
- Mulvale GM, Nguyen TD, Miatello AM, Embrett MG, Wakefield PA, Randall GE. Lost in transition or translation? Care philosophies and transitions between child and youth and adult mental health services: a systematic review. J Ment Health. 2019;28(4):1–10.
- Paul M, Ford T, Kramer T, Islam Z, Harley K, Singh SP. Transfers and transitions between child and adult mental health services. Br J Psychiatry Suppl. 2013;54:s36–40.

- Blum RG, D Hodgman C, et al. Transition from childcentered to adult health-care systems for adolescents with chronic conditions. J Adolesc Health. 1993;14:570–6.
- Garland BH, Caldwell KL, Acosta AB, Wiemann CM, Gonzales SA, Wolfe RS. Clinical considerations for emerging adults with eating disorders and transition to adult-based care. Evid Based Pract Child Adolesc Mental Health. 2018:1–15.
- Hergenroeder AC, Wiemann CM, Cohen MB. Current issues in transitioning from pediatric to adult-based care for youth with chronic health care needs. J Pediatr. 2015;167(6):1196–201.
- Singh S, Paul M, Ford T, Kramer T, Weaver T, McLaren S, et al. Lost in transition: a multi-perspective study of process, outcome and experience of transition from child to adult mental health care (TRACK). Br J Psychiatry. 2010;197:305–12.
- Pinhas L, Morris A, Crosby RD, Katzman DK. Incidence and age-specific presentation of restrictive eating disorders in children: a Canadian Paediatric surveillance program study. Arch Pediatr Adolesc Med. 2011; 165(10):895–9.
- Herpertz-Dahlmann B. Adolescent eating disorders: definitions, symptomatology, epidemiology and comorbidity. Child Adolesc Psychiatr Clin N Am. 2009;18(1):31–47.
- Arcelus J, Bouman WP, Morgan JF. Treating young people with eating disorders: transition from child mental health to specialist adult eating disorder services. Eur Eat Disord Rev. 2008;16(1):30–6.
- Franko DL, Keshaviah A, Eddy KT, Krishna M, Davis MC, Keel PK, et al. A longitudinal investigation of mortality in anorexia nervosa and bulimia nervosa. Am J Psychiatr. 2013;170(8):917–25.
- Dimitropoulos G, Tran AF, Agarwal P, Sheffield B, Woodside B. Navigating the transition from pediatric to adult eating disorder programs: perspectives of service providers. Int J Eat Disord. 2012;45(6):759–67.
- Garland BH, Caldwell KL, Acosta AB, Wiemann CM, Gonzales SA, Wolfe RS. Clinical considerations for emerging adults with eating disorders and transition to adult-based care. Evid Based Pract Child Adolesc Mental Health. 2019;4(2):187–201.
- Winston AP, Paul M, Juanola-Borrat Y. The same but different? Treatment of anorexia nervosa in adolescents and adults. Eur Eat Disord Rev. 2012;20(2): 89–93.
- Dimitropoulos G, Toulany A, Herschman J, Kovacs A, Steinegger C, Bardsley J, et al. A qualitative study on the experiences of young adults with eating disorders transferring from pediatric to adult care. Eat Disord. 2015;23(2): 144–62.
- Dimitropoulos G, Tran AF, Agarwal P, Sheffield B, Woodside B. Challenges in making the transition between pediatric and adult eating disorder programs: a qualitative study from the perspective of service providers. Eat Disord. 2013;21(1):1–15.
- Signorini G, Singh SP, Marsanic VB, Dieleman G, Dodig-Curkovic K, Franic T, et al. The interface between child/adolescent and adult mental health services: results from a European 28-country survey. Eur Child Adolesc Psychiatry. 2018;27(4):501–11.
- 20. Lockertsen V, Nilsen L, Holm W, Lill A, Rø Ø, Burger LM, Røssberg Jl. Mental health professionals' experience with the transition between the child and adolescent mental health system and the adult mental health system for patients with anorexia nervosa. Under review.
- Signorini G, Singh SP, Boricevic-Marsanic V, Dieleman G, Dodig-Ćurković K, Franic T, et al. Architecture and functioning of child and adolescent mental health services: a 28-country survey in Europe. Lancet Psychiatry. 2017;4(9): 715–24.
- Obeid N, McVey G, Seale E, Preskow W, Norris ML. Cocreating research priorities for anorexia nervosa: the Canadian eating disorder priority setting partnership. Int J Eat Disord. 2020;53(5):392–402.
- Hart LM, Wade T. Identifying research priorities in eating disorders: a Delphi study building consensus across clinicians, researchers, consumers, and carers in Australia. Int J Eat Disord. 2020;53(1):31–40.
- Anastasiadou D, Medina-Pradas C, Sepulveda A, Treasure J. A Systematic Review of Family Caregiving in Eating Disorders. Eat Behav. 2014;15(3): 464–77.
- 25. Stebbins RA. Exploratory research in the social sciences: Sage; 2001.
- Wilkinson S. Focus groups. In: Smith JA, editor. Qual Psychol. London: SAGE Publications Ltd; 2015.
- 27. Hummelvoll JK. Kunnskapsdannelse i praksis: handlingsorientert forskningssamarbeid i akuttpsykiatrien [developing knowledge in practice:

Co-operative inquery in acut psychiatry units]. Oslo: Universitetsforl; 2003. p. 286. s. fig. p.

- Malterud K. Systematic text condensation: a strategy for qualitative analysis. Scand J Public Health. 2012;40(8):795–805.
- 29. Giorgi A. Sketch of a psychological phenomenological method. Phenomenol Psychol Res. 1985:8–22.
- Newlove-Delgado T, Ford TJ, Stein K, Garside R. 'You're 18 now, goodbye': the experiences of young people with attention deficit hyperactivity disorder of the transition from child to adult services. Emot Behav Diffic. 2018;23(3):296–309.
- Cleverley K, Lenters L, McCann E. "Objectively terrifying": a qualitative study of youth's experiences of transitions out of child and adolescent mental health services at age 18. BMC Psychiatry. 2020;20(1):147.
- Nordbø RH, Espeset EM, Gulliksen KS, Skårderud F, Geller J, Holte A. Reluctance to recover in anorexia nervosa. Eur Eat Disord Rev. 2012; 20(1):60–7.
- Sly R, Morgan JF, Mountford VA, Sawer F, Evans C, Lacey JH. Rules of engagement: qualitative experiences of therapeutic alliance when receiving in-patient treatment for anorexia nervosa. Eat Disord. 2014;22(3):233–43.
- Oyer L, O'Halloran MS, Christoe-Frazier L. Understanding the working alliance with clients diagnosed with anorexia nervosa. Eat Disord. 2016;24(2): 121–37.
- 35. Vitousek K, Watson S, Wilson GT. Enhancing motivation for change in treatment-resistant eating disorders. Clin Psychol Rev. 1998;18(4):391–420.
- Swift JK, Greenberg RP. Premature discontinuation in adult psychotherapy: a meta-analysis. J Consult Clin Psychol. 2012;80(4):547–59.
- Eating disorders: Recognition and treatment. Full guideline [Internet]. 2017. Available from: https://www.nice.org.uk/guidance/ng69/evidence/fullguideline-pdf-161214767896. Accessed 13 Feb 2019.
- Appleton R, Connell C, Fairclough E, Tuomainen H, Singh SP. Outcomes of young people who reach the transition boundary of child and adolescent mental health services: a systematic review. Eur Child Adolesc Psychiatry. 2019:1–16.
- Sawyer SM, Azzopardi PS, Wickremarathne D, Patton GC. The age of adolescence. Lancet Child Adolesc Health. 2018;2(3):223–8.
- Scott E, Naismith S, Whitwell B, Hamilton B, Chudleigh C, Hickie I. Delivering youth-specific mental health services: the advantages of a collaborative, multi-disciplinary system. Australas Psychiatry. 2009;17(3):189–94.
- Malla A, Iyer S, Shah J, Joober R, Boksa P, Lal S, et al. Canadian response to need for transformation of youth mental health services: ACCESS open minds (Esprits ouverts). Early Interv Psychiatry. 2019;13(3): 697–706.
- 42. Hickie IB. Youth mental health: we know where we are and we can now say where we need to go next. Early Interv Psychiatry. 2011;5(s1):63–9.
- 43. Raballo A, Poletti M, McGorry P. Architecture of change: rethinking child and adolescent mental health. Lancet Psychiatry. 2017;4:656–8.
- Tierney S. The individual within a condition: a qualitative study of young People's reflections on being treated for anorexia nervosa. J Am Psychiatric Nurs Assoc. 2008;13(6):368–75.
- Offord A, Turner H, Cooper M. Adolescent inpatient treatment for anorexia nervosa: a qualitative study exploring young adults' retrospective views of treatment and discharge. Eur Eat Disord Rev. 2006;14(6):377–87.
- Malson H, Finn D, Treasure J, Clarke S, Anderson G. Constructing 'the eating disordered Patient'1: a discourse analysis of accounts of treatment experiences. J Community Appl Soc Psychol. 2004;14(6):473–89.
- Button EJ, Warren RL. Living with anorexia nervosa: the experience of a cohort of sufferers from anorexia nervosa 7.5 years after initial presentation to a specialized eating disorders service. Eur Eat Disord Rev. 2001;9(2):74–96.
- 48. Colton A, Pistrang N. Adolescents' experiences of inpatient treatment for anorexia nervosa. Eur Eat Disord Rev. 2004;12(5):307–16.
- Colli A, Speranza AM, Lingiardi V, Gentile D, Nassisi V, Hilsenroth MJ. Eating disorders and therapist emotional responses. J Nerv Ment Dis. 2015;203(11): 843–9.
- Werbart A, Andersson H, Sandell R. Dropout revisited: patient- and therapistinitiated discontinuation of psychotherapy as a function of organizational instability. Psychother Res. 2014;24(6):724–37.
- Vostanis P. Patients as parents and young people approaching adulthood: how should we manage the interface between mental health services for young people and adults? Curr Opin Psychiatry. 2005;18(4):449–54.
- 52. Mjøsund NH, Eriksson M, Espnes GA, Haaland-Øverby M, Jensen SL, Norheim I, et al. Service user involvement enhanced the research quality in

a study using interpretative phenomenological analysis-the power of multiple perspectives. J Adv Nurs. 2017;73(1):265–78.

- Gillard S, Simons L, Turner K, Lucock M, Edwards C. Patient and public involvement in the coproduction of knowledge: reflection on the analysis of qualitative data in a mental health study. Qual Health Res. 2012;22(8): 1126–37.
- Juritzen TI, Soberg HL, Roe C, Saebu M, Engen G, Bliksvaer T, et al. The one or the many: quantified subjectivity and aggregated uniqueness in qualitative rehabilitation research. Qual Health Res. 2017;27(1):51–9.

### **Publisher's Note**

Springer Nature remains neutral with regard to jurisdictional claims in published maps and institutional affiliations.

### Ready to submit your research? Choose BMC and benefit from:

- fast, convenient online submission
- thorough peer review by experienced researchers in your field
- rapid publication on acceptance
- support for research data, including large and complex data types
- gold Open Access which fosters wider collaboration and increased citations
- maximum visibility for your research: over 100M website views per year

### At BMC, research is always in progress.

Learn more biomedcentral.com/submissions

